Author's Response To Reviewer Comments

Clo<u>s</u>e

We have addressed suggested editorial points:

Supplemental material is updated.
biotools ID is updated.
GigaDB dataset is added in references and cited from "data availability" paragraph.

Clo<u>s</u>e